Author's response to reviews

Title: Amniotic fluid embolism incidence, risk factors and outcomes: a review and recommendations

Authors:

Marian Knight (marian.knight@npeu.ox.ac.uk)
Cynthia Berg (cjb3@cdc.gov)
Peter Brocklehurst (peter.brocklehurst@npeu.ox.ac.uk)
Michael Kramer (michael.kramer@mcgill.ca)
Gwyneth Lewis (drgwyneth.lewis@gmail.com)
Jeremy Oats (jeremy.oats@rwh.org.au)
Christine L Roberts (clroberts@med.usyd.edu.au)
Cathy Spong (spongc@dir49.nih.gov)
Elizabeth Sullivan (E.Sullivan@unsw.edu.au)
Jos van Roosmalen (J.J.M.van Roosmalen@lumc.nl)
Joost Zwart (J.J.Zwart@lumc.nl)

Version: 6 Date: 30 January 2012

Author's response to reviews:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)
We have revised the list as requested.

Abstract: Please structure this using the following sub-headings only:
Background, Methods, Results, Conclusions
We have made the requested change.

Tables: tables 2, 3 and 4 have been divided into parts. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber these tables and any references to them in the text. Alternatively, please merge each set of tables into one larger table.
We have merged each set of tables into a larger table as suggested.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript, and change any coloured text to black
We have removed all highlighting and coloured text.
Please rename "Contributions" to "Authors' Contributions"
We have made this change

Tables: Please remove all visible vertical lines from your tables.
We have made this change